Author's response to reviews

Title: Do children with cerebral palsy benefit from computerized working memory training?: study protocol for a randomized controlled trial.

Authors:

Gro CC Løhaugen Dr (gro.c.lohaugen@ntnu.no)
Harald Beneventi Dr (harald.beneventi@sus.no)
Guro Andersen Dr MD (guro.andersen@siv.no)
Cato Sundberg physiotherapist (cato.sundberg@sshf.no)
Heidi F Østgård psychologist (heidi.f.ostgard@ntnu.no)
Ellen Bakkan special educator (ellen.elise.bakkan@sus.no)
Geir Walther psychologist (geir.walther@siv.no)
Torstein Vik Professor MD (torstein.vik@ntnu.no)
Jon Skranes Professor MD (jon.skranes@ntnu.no)

Version: 2 Date: 16 May 2014

Author's response to reviews:

Documentation regarding ethical approval has now been included both in English and Norwegian as requested. Documentation regarding funding has been included in English.

Please ensure the title conforms to journal style for study protocol articles. The title should follow the format “____________: study protocol for a randomized controlled trial.” Response: This has now been altered accordingly.

2. Please indicate the corresponding author on your title page. Response: This is now included on the title page.

3. Please include the date your study was registered with your trial registration number at the end of your Abstract. Response: The study was registered in April 2014 and this information is now added.

4. Please include a statement in your Methods section explaining that you obtained informed consent from each participant. Response: This has been included.

5. Please include an acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does
not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

Response: Thank you for this opportunity, and this information is included as suggested.

6. Please move your list of abbreviations above your Competing Interests section. Response: This section is moved as suggested.